Author's response to reviews

Title: Comparison of primary care experiences among adults in general outpatient clinics and private general practice clinics in Hong Kong

Authors:

Samuel YS Wong (yeungshanwong@cuhk.edu.hk)
Kenny Kung (kungkl@ha.org.hk)
Sian M Griffiths (siangriffiths@cuhk.edu.hk)
Tanya Carthy (tanya_carthy@hotmail.com)
Martin CS Wong (wong_martin@cuhk.edu.hk)
Su V Lo (losv@ha.org.hk)
Vincent CH Chung (vchung@cuhk.edu.hk)
William B Goggins (wgoggins@cuhk.edu.hk)
Barbara Starfield (bstarfie@jhsph.edu)

Version: 8 Date: 29 June 2010

Author's response to reviews: see over
Dear Dr. Norton,

Ref: Comparison of primary care experiences among adults in general outpatient clinics and private general practice clinics in Hong Kong

Thank you for accepting our manuscript for publication in BMC Public Health. We have made further improvement for our manuscript regarding the format change request of the manuscript. The details of our response is attached point by point in the following page. We believe our findings will be of great interest to public health generally, and particularly to researchers interested in the primary healthcare services. As the premier international journal devoted to publish articles in the understanding of all aspects of public health, BMC Public Health represents the perfect platform for us to share these results with the international research community.

We confirm that the manuscript has not been published elsewhere and is not under consideration by another journal. All authors have approved the manuscript and agreed with submission to BMC Public Health. There are no competing interests to declare.

Thank you very much for your attention. We shall look forward to hearing from you at your earliest convenience.

Sincerely,

Samuel YS Wong.
Associate Professor,
The Chinese University of Hong Kong.
Major revisions

1. Author list: the list of authors in the manuscript should be written exactly as it is in the submission system, both in style and order. The preferred style is ?First name Initial Last name? (e.g. Joe F Bloggs).

A: We have revised the author list according to suggested style.

2. Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

A: Again, affiliations now conform to the journal style.

Minor Revision

1. References: Please check if reference 27 'In Press' has now been published. If so, please provide full citation details.

A: Reference 27 has now been published. We have updated our reference list accordingly.